Author's response to reviews

Title: Parents' reported preference scores for childhood atopic dermatitis disease states

Authors:

Joelle Y Friedman (joelle.friedman@duke.edu)
Shelby D Reed (shelby.reed@duke.edu)
Kevin P Weinfurt (kevin.weinfurt@duke.edu)
Kristijan H Kahler (kristijan.kahler@pharma.novartis.com)
Emmanuel B Walter (walte002@mc.duke.edu)
Kevin A Schulman (kevin.schulman@duke.edu)

Version: 4 Date: 15 October 2004

Author’s response to reviews: see over
October 15, 2004

BMC Pediatrics
The BioMedCentral Editorial Team
Tel: +44 (0)20 7631 9921
Facsimile: +44 (0)20 7636 4296
e-mail: editorial@biomedcentral.com

RE: Revisions to Manuscript 7292816902964883 for BMC Pediatrics

To Whom It May Concern:

We have made the following revisions to our manuscript:

1. Removed unnecessary capitalization from the title, section headings and subheadings, and figure and table titles.
2. Changed reference list to conform to journal style.
3. Verified and made corrections to references.
4. Cropped figure.

My coauthors and I thank you for your consideration of our manuscript. Please contact me if additional revisions are needed.

Sincerely,

Kevin A. Schulman, MD

Encl.